
• Thu Dec 22nd & Fri Dec 23rd: I:KPS6N & I:KPS6S extraction kickers went into alarm: 

elog230315, elog230397 & elog230399

– May be cold weather related

• Fri Dec 23rd & Wed Dec 28th: Low transformer oil level status for E:V108: elog230346, 

elog230379, elog230399, & elog230675

– Status is now cleared; sounds like it was cold weather related

– Thanks to EE Support

• Sun Dec 25th: E:QD118 was spraying water: elog230477 & elog230659

– Thanks to Ops for finding and stopping the spraying water

– Experts will meet next week to form a repair plan

• Mon Dec 26th to Fri Dec 30th: Horn & stripline work continues

– Let experts describe work
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https://www-bd.fnal.gov/Elog/?orEntryId=230315
https://www-bd.fnal.gov/Elog/?orEntryId=230397
https://www-bd.fnal.gov/Elog/?orEntryId=230399
https://www-bd.fnal.gov/Elog/?orEntryId=230346
https://www-bd.fnal.gov/Elog/?orEntryId=230379
https://www-bd.fnal.gov/Elog/?orEntryId=230399
https://www-bd.fnal.gov/Elog/?orEntryId=230675
https://www-bd.fnal.gov/Elog/?orEntryId=230477
https://www-bd.fnal.gov/Elog/?orEntryId=230659

